Author’s response to reviews

Title: The development of a new corporate specific health risk measurement instrument, and its use in investigating the relationship between health and well-being and employee productivity

Authors:

Peter R Mills (p.mills@vielife.com)

Version: 5 Date: 27 January 2005

Author’s response to reviews:

1. Address for correspondence moved to the beginning of the affiliation
2. Table 1 moved to after the reference list and changed to portrait dimensions.